Author's response to reviews

Title: Wireless LAN in Pre-hospital Environment

Authors:

Dongquan Chen (dongquan.chen@ccc.uab.edu)
Seng-Jaw Soong (sjssoong@uab.edu)
Gary J Grimes (ggrimes@uab.edu)
Helmuth F Orthner (horthner@uab.edu)

Version: 6 Date: 19 August 2004

Author's response to reviews:

Dear Editors,

I have uploaded the revision of the manuscript, which has been copyedited through sciencemanger.com. The following modifications as suggested by you and the copyeditor have been made:

1. Most, if not all, changes made by copyeditor have been accepted.
2. All credentials (PhD, et al.,) have been removed.
3. The email addresses of all co-authors have been added to the title page.
4. Corresponding author is indicated.
5. Numbers in all tables are rounded up.
6. All IP addresses in the figures are removed.
7. Length measurement has been changed from foot/feet into meters.
8. Reference has been reedited to include all authors.

Please let me know if you need further information. I look forward to the publication of the paper. Thank you.

Dongquan